Author's response to reviews

Title: The prevention of anaphylactoid reactions to iodinated radiological contrast media: A systematic review of randomised controlled trials

Authors:

Anthony Delaney (adelaney@med.usyd.edu.au)
Andrew Carter (andrewccarter@yahoo.com.au)
Malcolm Fisher (mfisher@med.usyd.edu.au)

Version: 3 Date: 25 April 2006

Author's response to reviews:

Anthony Delaney MBBS MSc FACEM FJFICM
Staff Specialist in Intensive Care
Royal North Shore Hospital
Pacific Highway
St. Leonards
NSW, 2065
Australia
Ph 61 2 9926 7111
Fax 61 2 9439 8418
Email adelaney@med.usyd.edu.au

Tuesday, April 25, 2006

Dear Biomed Central Editorial team and Reviewers.

Thank you for your consideration of our manuscript. Please find below a description of the changes that have been made to the manuscript.

1. In response to Dr Liu's comment, we have clarified the recommendations in the conclusion of the abstract to reflect more specifically the findings of the review. Specifically we have differentiated between the regimens that have been tested for ionic and non-ionic contrast.

The manuscript has also been reformatted to reflect the formatting requirements of BMC medical imaging. My apologies for not having this sorted out at an earlier stage. Specifically, we have:

* Removed the underlining from the title
* Removed the unnecessary capitalisation from the title.
* Removed the titles from the list of authors
* My current affiliations have been updated
* The highlighting from the revised manuscript has been removed
* The figure titles have been removed from the figures, and added to the manuscript after the References
* The figures have been cropped

I trust that this takes care of all the revisions and formatting changes. If there is anything else that needs attention, please don't hesitate to contact me.

Thank you for all of your time and consideration.

Sincerely,

Anthony Delaney